Course Information Sheet Calculus III (Math208), Section 004, Spring 2004, HENZ 201 11:30 p.m.–12:20 PM, M-W-R-F University of Nebraska – Lincoln

Instructor: Dr. Jon-Lark Kim. Office: 936 Oldfather Hall. Phone: 472-7253. E-mail: jlkim@math.unl.edu

http://www.math.unl.edu/∼jlkim

Office Hours: M-W-F 10:30–11:20 AM

Text: Calculus, Smith and Minton, Second Ed. McGraw-Hill, 2002.

Calculators: A graphing calculator is required for this course. The Texas Instruments TI-85 or TI-86 is recommended.

• Always bring your calculator to class! You will need it for practical work in class.

Homework and Reading: Beside each lecture date on the syllabus, you will find a list of sections from the text to be discussed in lecture and problems from the sections. Please read the text prior to class. You are to work ALL these problems (unless I say otherwise during lecture) before the next lecture. You learn this material by doing homework. So I strongly recommend that you hand in homework each Monday.

Quizzes: There will be a (10-15min) quiz at the end of every Friday class. It will cover previous materials and homework problems. I’ll announce which sections will be on the quiz before Friday. No quizzes on exam weeks! There might be a few pop quizzes.

Exams: There will be three midterm exams and one final exam.

Midterm Exams: Schedules of the midterms are found on the syllabus. The dates might chagne. I expect you to arrange your schedule to accommodate the exam dates or to choose another section of Math 208.

Final Exam: The final will be two hours in length and it will be given on Wednesday, May 5, from 6–8 p.m. The location of the final will be announced during the last week of class.

Grading Policy: The following table shows how points will be awarded over the course:

Number Pts. Total Midterm Exams 3 100 300 Quizzes highest 10 10 100 Homowork highest 10 5 50 Final Exam 1 200 200 Total 650

At the end of the course, your final grade will be based on the number of points you receive. While I may choose to adjust the course grading scale, the following table gives the minimum grades as a function of point totals:

Total Percentage (%) 93 90 87 83 80 77 73 70 67 63 60 57 below 57 Minimum Grade A+ A A− B+ B B− C+ C C− D+ D D− F